Author's response to reviews

Title: Females do not have more injury road accidents on Friday the 13th

Authors:

Igor Radun (igor.radun@helsinki.fi)
Heikki Summala (heikki.summala@helsinki.fi)

Version: 6 Date: 12 November 2004

Author's response to reviews:

Dear Editor,

thank you very much for your positive message. Please find enclosed our manuscript with final changes done as requested. The changes are listed below, as response to your requests:

Formatting changes requested

References - References should be cited in the text using consecutive numbers in square brackets.
CORRECTED. Note: Instead, the full citation in Abstract is not given in square brackets.

Figure legends - Please place the figure legends in the main text, after the references.
OK

Tables - Table 1 has been divided into a, b and c. It is important that tables are numbered in ascending numerical order - 1, 2 and 3. If you choose to renumber the tables in this way can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes. Each table will also require a title. Alternatively, you may wish to make the tables into one composite table.
Tables - Tables must be divided into cells / fields.

WE HAVE NUMBERED TABLES THROUGH 1-3, ALSO WHEN CITED IN TEXT, AND GIVEN A TITLE FOR EACH OF THEM

The borders of tables included as part of the manuscript file must be visible black lines. Tables generated with tabbed text are not acceptable.

OUR TABLES ARE WORD TABLES WITHOUT TABS, ALL THE NUMBERS ARE CENTERED IN CELLS (CAN BE CHANGES EASILY), AND BORDERS OF TABLES HAVE BEEN MADE WITH WORD FORMAT - BORDERS AND SHAPE OPERATION, ONLY HORIZONTAL BORDERS ARE WITH VISIBLE BLACK LINES ALTHOUGH IN WORD TABLE IT SEEMS AS IF ALL BORDERS WOULD BE MARKED. WE CHECKED THAT THIS IS NORMAL WAY OF BORDERING TABLES IN RECENT ACCEPTED PAPERS IN BMC Public Health.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.

FIGURE IS NOW ENCLOSED AS CROPPED

We have also done a few minor corrections in language (FOR EXAMPLE TITLE: DO NOT HAVE pro...
Please let us know if we have missed something.

We thank very much for the fast editorial process and remain waiting for seeing our work in the journal website.

With kind regards,

Heikki Summala and Igor Radun